Author’s response to reviews

Title: Colon Cancer Screening Among African American Church Members: A Qualitative and Quantitative Study Of Patient-Provider Communication

Authors:

Mira L Katz (katz-4@medctr.osu.edu)
Aimee James (ajames@kumc.edu)
Michael P Pignone (michael_pignone@med.unc.edu)
Marlyn Hudson (allicock@email.unc.edu)
Ethel Jackson (etheljeanj@aol.com)
Veronica Oates (voates@email.unc.edu)
Marci K Campbell (marci_campbell@unc.edu)

Version: Date: 10 October 2004

Author’s response to reviews:

October 10, 2004

Iratxe Puebla
Assistant Editor
BMC Public Health

Dear Iratxe Puebla,

We appreciate the thorough and thoughtful review of our manuscript Colorectal cancer screening among African American church members: a qualitative and quantitative study of patient-provider communication (MS: 5459783832391996)

Per the suggestion of the reviewer, a note has been added regarding the health care facility to Table 3. Additionally, requested format changes have been made to the manuscript.

Thank you,

Mira L. Katz, Ph.D., M.P.H.
Assistant Professor
School of Public Health
The Ohio State University
Columbus, Ohio